Author's response to reviews

Title: Perioperative oxygen fraction - effect on surgical site infection and pulmonary complications after abdominal surgery: a randomized clinical trial. Rationale and design of the PROXI-Trial

Authors:

Christian S Meyhoff (meyhoff@rh.dk)
Jørn Wetterslev (wetterslev@ctu.rh.dk)
Lars N Jorgensen (larsnjorgensen@hotmail.com)
Steen W Henneberg (steen.henneberg@rh.regionh.dk)
Inger Simonsen (i.s@dadlnet.dk)
Therese Pulawska (therese@pulawska.dk)
Line R Walker (walker@laege.dk)
Nina Skovgaard (nina.skovgaard@dadlnet.dk)
Kim Heltø (kimheltoe@gmail.com)
Peter Gocht-Jensen (peter_gocht@hotmail.com)
Palle S Carlsson (palle.carlsson@as.aaa.dk)
Henrik Rask (henrikrask@dadlnet.dk)
Sharaf Karim (sharaf73@hotmail.com)
Charlotte G Carlsen (charlotte@greenhesel.dk)
Frank S Jensen (frje@geh.regionh.dk)
Lars S Rasmussen (lsr@rh.dk)
PROXI Trial Group (not@vaid.com)

Version: 5 Date: 19 October 2008

Author's response to reviews:

References 43-46 were only cited in the tables. These references are now cited in the manuscript text on page 9 as references #27-30. The following references are updated accordingly.